Author's response to reviews

Title: fish bone foreign body presenting with acute fulminating retropharyngeal abscess in a resource challenged center: case report

Authors:

Olushola A Afolabi (droaafolabi@yahoo.com)
Joseph O Fadare (jofadare@gmail.com)
Ezekiel O Oyewole (droyewole2002@yahoo.com)
Stephen A Ogah (stephenogah@yahoo.com)

Version: 3 Date: 21 September 2010

Author's response to reviews:

21/09/2010
The editor,
Journal of medical Case report
Dear Sir

COVERING LETTER FOR THE MANUSCRIPT TITLED: FISH BONE FOREIGN BODY PRESENTING WITH ACUTE FULMINATING RETROPHARYNGEAL ABSCESS IN A RESOURCE CHALLENGE CENTER: CASE REPORT

The authors wish to inform the editor that the correction noted by both the associate editor and the editor has been addressed and all the changes made has been addressed in highlighted text.
The acronyms in the paper have been addressed as requested.
The relevant papers provided has been reviewed and added to the introduction, discussion and conclusion section of the paper.
Also a header has been inserted into the conclusion section as well as clear explanations as regards the relevance of the case report to the field of medicine have been addressed.
Thanking you in anticipation
Yours Faithfully,
Signed
Dr O.A. Afolabi
(Correspondence author)